Author's response to reviews

Title: Usability evaluation with adolescents and mental health professionals to develop a web-based cognitive-behaviour therapy program for adolescents with anxiety disorders

Authors:

  Lori M Wozney (lori.wozney@iwk.nshealth.ca)
  Pamela Baxter (BaxterP@mcmaster.ca)
  Amanda S Newtwon (an6@ualberta.ca)

Version:2 Date: 18 September 2014

Author's response to reviews:

Hello,

Line numbers and page numbers were added to the document as requested.

Sincerely,

Dr. Lori Wozney